Author's response to reviews

Title: Evaluation of psychological and physiological predictors of fatigue in patients with COPD

Authors:

Agnieszka Lewko (agnieszka_lewko@yahoo.co.uk)
Penelope L Bidgood (bidgood@kingston.ac.uk)
Rachel Garrod (rachel.garrod@nhs.net)

Version: 5 Date: 16 October 2009

Author's response to reviews:

Minor corrections:
1. Background: References [10,13] moved to the end of the sentence
2. Methods, subject recruitment: First line in the bracket (defined as..... with a general practitioner or chest physician diagnosis)- 'a' added after 'with' and deleted after 'or'
3. Results, First sentence: Deleted (52 male) and (12 male)
4. Discussion, page 11, line 11 from the bottom: Word 'should' added in 'Hence, the multi- rather than uni- dimensional assessment should be considered.'
5. Table 1, legend: one ; deleted after FFMI definition